Author's response to reviews

Title: A successful treatment of methemoglobinemia presenting in an elderly couple with focal seizures and severe cyanosis: a case series

Authors:

Ying-Fu Su (winesoul2003@yahoo.com.tw)
Li-Hua Lu (yelrihs6992@yahoo.com.tw)
Tai-Hao Hsu (th4420@gmail.com)
Shih-Liang Chang (slc0124@gmail.com)
Rong-Tsung Lin (lin530323@gmail.com)

Version: 6 Date: 21 February 2012

Author's response to reviews: see over
Title: A successful treatment of methemoglobinemia presenting in an elderly couple with focal seizures and severe cyanosis: a case series

Dear editor

Thank you for your precious comments and suggestions for the revision of the title of our manuscript. We have modified the title accordingly, and the detailed corrections are listed below point by point:

1) Please amend title to be more descriptive; adhering to the following format; 'A presenting in B with C: a case series.

We have revised the title, “A successful treatment of methemoglobinemia presenting in an elderly couple with focal seizures and severe cyanosis: a case series” to conform the format.

2) Please upload the figures as separate files via the online submission system. They should not be included within the main manuscript document.

We have uploaded the figures as separate files via the online submission system accordingly.

3) Please crop figure 1 (with patient’s face) to preserve the anonymity. Please ensure that only the feature relevant to the case report are shown.

We have cropped the figure 1 (without patient’s nose and eye) to preserve patient’s anonymity.

The cover letter and manuscript has been resubmitted to your journal. We look forward to your positive response. Thank you very much!

Yours Sincerely,

Rong-Tsung Lin, MD 2012/02/21

Tung’s Taichung Metro Harbor Hospital, Taichung

E-mail:lin530323@gmail.com